Dear editorial team,

I read the manuscript carefully and noticed two things I would like to point out:

1. Discussion, the first paragraph
You have edited the last sentence as follows:
"However, the concentration of our patient’s dexmedetomidine greatly varied even during unchanged infusion."

I am not good enough in English to understand minor details and differences. Does this mean that the infusion rate was constant but the dexmedetomidine concentration in plasma varied? This is what I am trying to say.

2. Table 1
During editing the manuscript, the abbreviation conc had been replaced by word “concentrate”. That is wrong, it should be “concentration”. Because the abbreviation is no longer used, I deleted it from the footer.

Kind regards,

Timo Iirola